Author's response to reviews

Title: A study of cytokeratin 20 immunostaining in the urothelium of neuropathic bladder of patients with spinal cord injury

Authors:

Subramanian Vaidyanathan (vaidyanathansiu@hotmail.com)
Ian W McDicken (IW.McDicken@mail.rlbuf.nwest.nhs.uk)
Anna J Ikin (AJ.Ikin@mail.rlbuf.nwest.nhs.uk)
Paul Mansour (Paul.Mansour@mail.soh-tr.nwest.nhs.uk)
Bakul M Soni (Bakul.Soni@mail.soh-tr.nwest.nhs.uk)
Gurpreet Singh (Gurpreet.Singh@mail.soh-tr.nwest.nhs.uk)
Pradipkumar Sett (Pradip.Sett@mail.soh-tr.nwest.nhs.uk)

Version: 2 Date: 20 Jul 2002

Dear Editor

I thank you for considering our submission for publication in your esteemed journal. We revised the manuscript as advised by the reviewers. The title has been changed. Results of immunostaining in the metaplastic squamous cells is given as advised by Dr Jian Yu Rao. Groups 1 and 2 are omitted. Figures 1 and 2 were omitted. Additional references have been provided as advised by Professor Southgate. Tables 1 and 2 were omitted. Now the manuscript focuses on spinal cord injury patients. Acknowledgement section has been rewritten. Discussion has been modified as per the advice of Professor Southgate. All other suggestions made by Professor Southgate have been implemented.

I thank Dr Jian Yu Rao and Professor Southgate for their valuable comments.

Yours sincerely

Dr S Vaidyanathan